Author’s response to reviews

Title: Encapsulating peritoneal sclerosis in a patient after allogeneic hematopoietic stem cell transplantation

Authors:

Yoshimitsu Shimomura (shimomura_0119@yahoo.co.jp)
Shinsuke Sakai (sakai@kid.med.osaka-u.ac.jp)
Hiroyuki Ueda (hioueda@kcho.jp)
Kohei Fujikura (kofujikura@gmail.com)
Yukihiro Imai (yukiimai@kcho.jp)
Takayuki Ishikawa (ishikawa@kcho.jp)

Version: 1 Date: 21 Oct 2017

Author’s response to reviews:

Responses to the Comments

Thank you for the careful review. We corrected the points you pointed out. Our responses to all the comments from the editor and reviewers are included below.

Comments:

1. Please provide a clearer statement in the ‘Ethics and consent to participate’ and in the ‘consent for publication’ sections regarding how you obtained the informed consent from the patient and/or his family, given that the patient died before the submission of the manuscript.

Thank you for the comments. We added clearer statement of consent to participate.

<Revised>

Page 13, Lines 184-186

Written informed consent was obtained from the patient at onset of EPS. In addition, the patient’s family gave permission for an immediate autopsy and publication after the patient died.
Written informed consent for publication of the case report and any accompanying images was obtained from the patient at onset of EPS. In addition, the patient’s family gave permission for publication at autopsy.

2. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Thank you for the comments. We added the effect that all authors have read and approved the manuscript.

<Added>

Page 13, Line 202

All authors have read and agree to the manuscript as written.

3. Please state whether there is any non-financial conflict of interest in the Competing Interests section.

Thank you for the comments. We added non-financial conflict of interest.

<Revised>

Page 13, Line 196

The authors declare no competing financial and non-financial interests.